<mark>,</mark>



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL103925

Key Points:

- If sediment dispersal changes through time and time is interpolated from long-term rates, one will misestimate durations from strata
- Misestimation is not inevitable in the strata. Some changes in sediment dispersal result in distortion, while others do not
- Geologic clues that indicate the degree of sediment localization are key to diagnosing stratigraphic sections with substantial distortion

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

E. A. Barefoot, eabarefoot@gmail.com

Citation:

Barefoot, E. A., Nittrouer, J. A., & Straub, K. M. (2023). Sedimentary processes and the temporal resolution of sedimentary strata. *Geophysical Research Letters*, 50, e2023GL103925. https://doi. org/10.1029/2023GL103925

Received 29 MAR 2023 Accepted 5 JUN 2023

Author Contributions:

Conceptualization: Eric A. Barefoot, Kyle M. Straub Funding acquisition: Eric A. Barefoot, Kyle M. Straub Investigation: Eric A. Barefoot Methodology: Eric A. Barefoot, Kyle M. Straub Resources: Kyle M. Straub Software: Eric A. Barefoot Supervision: Jeffrey A. Nittrouer, Kyle M. Straub Validation: Eric A. Barefoot, Kyle M. Straub

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Sedimentary Processes and the Temporal Resolution of Sedimentary Strata

Eric A. Barefoot¹, Jeffrey A. Nittrouer², and Kyle M. Straub³

¹Department of Earth and Atmospheric Sciences, Indiana University, Bloomington, IN, USA, ²Department of Geosciences, Texas Tech University, Lubbock, TX, USA, ³Earth and Environmental Sciences Department, Tulane University, New Orleans, LA, USA

Abstract An appealing strategy for reconstructing the timing and tempo of paleoenvironmental change from sedimentary strata is to linearly interpolate between marker beds of known age. This method requires significant assumptions, but more advanced age modeling methods are usually not feasible. We used experiments to explore how changes in sedimentary processes invalidate these assumptions and affect estimates of time from the strata. When sedimentary processes changed to favor widespread deposition, we found that measuring time linearly systematically overestimated time duration from the resulting strata (time dilation) and misestimated the beginning and end of geologic intervals (phase shifting). When simple age models must be used for sedimentary strata, geologic evidence for transient changes in spatial sediment dispersal may help identify sections of dilated and shifted time, and better resolve time in sedimentary strata.

Plain Language Summary Sedimentary rocks can be used to measure the amount of time that passed between events in Earth history. Since it is rare to find layers of rock whose age can be measured, one needs a simplifying assumption: the thickness of sedimentary rocks is proportional to the amount of time that passed. In other words, the sediment accumulated relatively steadily, and gaps are short relative to the duration of interest. In this study, we show that this approach incorrectly dates events in the timeline, and erroneously inflates time durations. The bias is most significant when a change in environmental conditions causes sediment to spread more widely across the landscape than it did before. The bias is not as pronounced when sediment deposits are localized. Clues and context from sedimentary rocks may help reveal whether sediment was spreading widely or locally in the past. Using this evidence, geologists will be able to identify sections where simplifying assumptions will not hold, and can apply appropriate corrections.

1. Introduction

Sedimentary rocks are the main geologic archive used to reconstruct paleoclimate, tectonic motions, and the history of life on Earth (Paola et al., 2018). Measuring the magnitude and duration of environmental change is crucial for understanding how Earth systems interact to maintain a habitable planet (Kirtland Turner, 2018). The most commonly-used age models assume relatively steady, continuous sedimentation to interpolate linearly between marker beds of known age. That is, gaps should be short compared to the interval of interest, and distributed uniformly in the stratigraphic column. When these assumptions do not hold, establishing a timeline from sedimentary strata is not straightforward.

Issues of timing and duration are central to interpreting many consequential events in Earth's history. For example, explaining the cause of the Cretaceous-Paleogene (K-Pg; ~65 Ma) mass extinction (Schoene et al., 2019) depends on the relative timing of the Deccan Traps eruptions (Keller et al., 2009; Schoene et al., 2015) versus the Chicxulub impact (Schulte et al., 2010; Sharpton et al., 1992). In another case, estimates for the duration and tempo of the Paleocene-Eocene Thermal Maximum (PETM; ~55.6 Ma) (Kirtland Turner, 2018) are essential for understanding how the global climate responded to past changes in the carbon cycle (Zachos et al., 1993). As a third example, the shape, duration, and magnitude of the Shuram carbon isotope excursion (CIE) at ca. 570 Ma are very challenging to constrain (Knoll et al., 1986; Le Guerroué et al., 2006), and yet the CIE is widely used as a chronostratigraphic marker to measure the timing of major evolutionary, tectonic, and climatic changes occurring in the Ediacaran period (Fike et al., 2006; Halverson et al., 2005; Williams & Schmidt, 2018).

The fundamental limit to overcoming these uncertainties and reconstructing geologic time in sedimentary rocks is the temporal incompleteness of strata (Sadler, 1981; Trampush & Hajek, 2017). Sedimentary strata



contain unconformities due to both external (allogenic) factors and internal (autogenic) unsteadiness (Davies & Shillito, 2018; Jerolmack & Sadler, 2007; Rogers, 1998).

We use the term "unconformity" throughout to refer to gaps in the stratigraphic record of any duration, due to any cause. Allogenic changes in environmental boundary conditions (e.g., sea level, tectonic uplift) that erode or starve basins of sediment leave unconformities in the stratigraphic column. Within a geologic basin, emergent internal (i.e., autogenic) sediment transport dynamics focus erosion and deposition in some portions of the basin while other regions receive no sediment at all (Davies & Shillito, 2021; Ganti et al., 2011; Straub & Foreman, 2018; Straub et al., 2009). As a result, for a given stratigraphic column, geologic time does not advance up section linearly. Instead, periods of inactivity and erosion are interspersed with bursts of sediment accumulation (Paola et al., 2018; Schumer et al., 2011; Tipper, 2015).

Layers of known age are uncommon, which makes it challenging to calibrate age models to account for sections of missing time in sedimentary strata. In the absence of other information, the default approach is to interpolate linearly between layers of known age, which inherently assumes that the sediment accumulation rate between them is steady and constant (Abels et al., 2010; Jarochowska et al., 2020; Westerhold et al., 2009; Wilf et al., 2003). This approximation is best applied to environments where the short-term variability of sediment accumulation is small compared to the long-term accumulation rate, such as lacustrine or abyssal sediments (exemplified in Figure 1—light blue line) (Foreman & Straub, 2017; Trampush & Hajek, 2017). While lakes and the deep oceans are important archives for tracing the behavior of Earth systems, they represent a subset of the available sedimentary rock archive. Important insights can be gained if paleoenvironmental interpretations with improved timescales are applied to terrestrial and marginal marine environments with highly variable sedimentation rates. In these environments, sediment accumulation rates fluctuate according to both the processes internal to the system itself (autogenic), as well as externally-imposed (allogenic) environmental conditions like sediment supply (Bryant et al., 2018; Syvitski & Saito, 2007), and hydrograph variability (Colombera et al., 2017; Esposito et al., 2018; Fielding et al., 2018; Leary & Ganti, 2020).

There are a variety of age-modeling approaches that exploit the signatures of allogenically-forced increases or decreases in sedimentation rate to sharpen age models. They include: (a) incorporating lithological variation to better infer accumulation rates (Kominz & Bond, 1990; Trauth, 2014), (b) using visual correlation (Zachos et al., 2005) and dynamic time warping algorithms to align records from disparate locations (Ajayi et al., 2020; Hay et al., 2019; Lisiecki & Lisiecki, 2002), and/or (c) aligning records with orbital cyclicities (Aswasereelert et al., 2013; Aziz et al., 2008; Meyers & Sageman, 2007; Röhl et al., 2007). These approaches, though, assume that sedimentation is continuous across large sections of the stratigraphic column, and assume unconformities are long-lived and rare. Moreover, because the stochasticity of sediment accumulation changes with allogenic forcing, the number and concentration of undetected unconformities may also covary in tandem with the environmental conditions (Figure 1—bottom row).

Because we lack the ability to directly measure the amount of unrecorded time in a stratigraphic column, assumptions like steady, continuous sedimentation will remain a practical means for estimating the timing and tempo of past climate, tectonics, and biological evolution. However, it is essential (a) to quantify how these assumptions impact our reconstructions of geologic time, and (b) to investigate how their influence can be mitigated.

2. Experiment Design

Herein, we use idealized sedimentary experiments with high-resolution age control to examine how changes in environmental boundary conditions affect reconstructions of geologic history when the linear time assumption does not hold. We use the term "event" throughout this paper to mean any interval of time in Earth's history whose beginning and end can be demarcated by evidence within the strata. Examples could include the interval between the first and last occurrence of index fossils, an isotope excursion measured from benthic foraminifera, or a shift from oxic to anoxic seawater.

To represent a generic sedimentary environment with well-characterized dynamics, we used an experimental basin (Li et al., 2016) where sediment and water flow created channelized surfaces (Hoyal & Sheets, 2009), and imposed net aggradation by steadily raising the water level in the basin over the full duration of the experiment. Throughout, we held the long-term average water and sediment supply constant, but imposed floods of varying





Figure 1. Sediment accumulation histories for four hypothetical scenarios. The left-hand panels show the coefficient of variation for sedimentation; the sediment accumulation histories are shown in the right-hand panels. The resulting stratigraphic columns are shown on the far right of each row. On the top row, the sedimentation variability for each scenario is a constant function of time (blue lines), whereas on the bottom row, the variability either increases or decreases during a geologic event (green lines). The beds in the resulting strata that were deposited during the climate event are highlighted in each column. Existing theory demonstrates that—when averaged over an ensemble of columns—estimates of timing and duration are unbiased for the cases in blue (Foreman & Straub, 2017; Trampush & Hajek, 2017). If steady sediment accumulation is assumed between dated horizons located at the top and bottom of each stratigraphic column, the perceived duration of the event in the dark green column would be compressed relative to its true duration, and the event in the light green column would be dilated. Either way, the result is unreliable estimates of the timing and duration of the event.

amplitudes. The flood amplitude (Q_{ν}) was defined as the ratio of the maximum discharge (Q_{max}) to the minimum discharge (Q_{min}) , $Q_{\nu} = Q_{\text{max}}/Q_{\text{min}}$, such that the average water discharge was equivalent for different values of Q_{ν} (see supplement for details). We controlled the spatial and temporal variability of sedimentation in the basin by engineering transitions between low- and high-amplitude flood regimes.

In these experiments, floods dispersed sediment across the surface; flood amplitude controlled how far away from a channel it dispersed. Under the high-amplitude flood regime ($Q_v = 3$), two main processes spread sediment widely across the surface. First, vigorous sheet flow during floods transported sediment away from channels and across the basin surface, depositing extensive channel-margin deposits. Second, high-amplitude floods promoted more frequent avulsions and accelerated channel migration (Barefoot, Nittrouer, & Straub, 2021; Esposito et al., 2018). In contrast, sediment dispersal under a low-amplitude flood regime ($Q_v = 1.5$) tended to be localized. During low-amplitude floods, sluggish overbank flows were incapable of transporting sediment far away from channels. This reduced channel mobility and focused sediment accumulation in the region adjacent to the channel (Barefoot, Nittrouer, & Straub, 2021).

We engineered transitions between the two flood regimes. Flood amplitude either diminished or intensified, and each transition took place over a prescribed duration. We call flood intensification "positive transitions" and diminishing floods "negative transitions." Each interval of positive or negative transition constitutes a geologic "event" whose duration and timing can be measured. Additionally, we compared the results to data from a control experiment, where flood amplitude remained unchanged throughout the entire experiment (Li et al., 2017; Li & Straub, 2020). For the control experiment, we chose an arbitrary window of a fixed duration to be the event of interest.

Throughout the experiment, we acquired topographic data and images of the sediment surface at high temporal frequency. Practically speaking, the smallest unconformity we can measure in our experiments spans the duration between two topographic scans. We located each event in the strata by identifying the stratigraphic horizons that marked the beginning and end of each event. For negative and positive flood transitions, the marker horizons of the event are the beginning and end of the transition. For the control experiment, the marker horizons were chosen so that events of the same duration could be compared to the flood transitions.

For each event, we picked a random set of stratigraphic columns from the experiment, simulating the process of taking a sediment core. We chose only those columns where the sediment thickness during the event was non-zero (see Supporting Information S1 for a discussion of this choice and its implications). The columns were sampled

at a regular spatial interval, much in the same way that one would sample a physical sediment core. The apparent duration of each event was measured using a linear time assumption (i.e., continuous sedimentation) between the bottom and top horizons of known age. We consider the apparent duration of the event as measured assuming linear time (t_m) relative to the true imposed duration (t_x) , and name the ratio of the two a "time dilation factor" $(t^* = t_m/t_x)$, where t^* values >1 represent apparent time dilation and t^* values <1 represent apparent time compression. We also measured the apparent timing of the event onset, and compared it to the value we prescribed. Again assuming linear time, we compute an apparent time lag (ϕ) , where values greater than zero indicate a lag, and values less than zero indicate a lead. For the remainder of the manuscript, we call this a "phase shift." We also use the general term "time distortion" to refer to the combined action of time dilation and phase shifting.

We normalized the imposed timing and duration of each event by a characteristic autogenic timescale. In the short run, accumulated sediment packages were uneven in thickness. Over the long run, the locus of deposition moved, and successively emplaced packages of sediment compensated for uneven short-term deposition (Ganti et al., 2011; Straub et al., 2009). Compensation occurs in nearly all net-depositional environments, and one may estimate a compensation timescale (T_c), defined as the ratio of the characteristic landscape relief (l) and the long-term aggradation rate (r); $T_c = l/r$ (Wang et al., 2011). This timescale is a property of the sedimentary dynamics; it reflects the time required for the largest and longest-lived autogenic sedimentation variability to average out. We use this timescale as a normalization factor because T_c is readily measurable in natural sedimentary systems, and allows a means to scale our experimental observations.

3. Process-Induced Apparent Time Dilation, Compression, and Phase Shifting

In all our experiments (positive, negative, and control), we observed that linear time interpolation generated apparent time dilation when the event of interest was short relative to the compensation timescale ($t_x = 1/4T_c$, see Figure 2). For intermediate length events ($t_x = 1/2T_c$), the reconstructed event time trended toward $t^* = 1$, indicating that the average perceived duration was equivalent to the true event duration (no time dilation). For the longest events ($t_x = T_c$), the reconstructed event time appeared shorter than the true duration ($t^* < 1$) indicating apparent time compression. In this way, the temporal sampling resolution of the stratigraphic record changed systematically with the duration of the geologic event itself. The dependence on apparent event duration rate increases as shorter time windows are measured. That is, we expect this dilation and compression simply because the stratigraphic column is incomplete, with many undetectable hiatuses.

There are differences between the control experiment and the experiments where we manipulated flood amplitude. Apparent time dilation in positive transitions (flood intensification) exceeded time dilation in the control experiment (>90% probability for all treatment levels, Figure 3). In other words, during time intervals when floods intensified, the resulting sediment package was thicker on average than a similar time interval when conditions did not change. If a linear age model were applied to reconstruct the duration of positive flood transition events, it would overestimate the true elapsed time by as much as 30%. This effect is above and beyond what would be expected due to the Sadler effect alone.

Our experiments also show an apparent time phase shift in the positive flood transitions. When using a linear time assumption, the apparent onset of the event in the strata leads the true onset by about $1/2T_c$ on average $(\phi \approx -1/2T_c)$; there is no clear trend with the event duration (Figure 2).

Both of these outcomes are asymmetric. Negative flood transition events are statistically indistinguishable from the control. When floods diminished in intensity, the apparent duration of the event is the same as corresponding events in the control experiment, and they do not lead or lag the true event onset (Figure 3).

4. Morphodynamic Controls on Stratigraphic Sampling

To explain why increasing flood amplitude created an apparent time dilation and phase shift in the stratigraphic record—but decreasing it had no effect—we examined how sedimentary processes responded to transient changes in flood amplitude.

In the absence of external forcing, our experimental landscapes converged on a dynamic equilibrium, where the roughness and morphology remain statistically similar over long intervals of time. The equilibrium topography formed by low-amplitude floods contrasts strongly with the equilibrium topography formed by high-amplitude



Geophysical Research Letters



Transition Sign - Positive - Negative - Control

Figure 2. Panel (a) shows the experimental events, with prescribed changes in flood amplitude in color. Example reconstructions using linear time interpolation are shown in gray for an ensemble of 25 randomly selected stratigraphic columns. An individual reconstruction is highlighted in black for each. Panel (b) shows the time dilation, t^* , calculated for the control experiment, as well as the experiment with positive and negative transitions. The mean is shown by a point, the thick horizontal line comprises 66% of the data, and the thin line 95% of the data. In general, the reconstructed duration of flood regime transitions is longer than the true value for abrupt transitions, and appears shorter for gradual transitions. Panel (c) shows the phase shift, ϕ , with the same convention. The phase shift observations indicates that event onset time leads the true onset for positive events, but not negative events or the control.

floods (Figure 4). Localized deposition during the low-amplitude flood regime constructed surfaces marked by short wavelength roughness features like ridges and levees. By widely dispersing sediment, high-amplitude flooding constructed smooth topography lacking ridges or swales (Barefoot, Nittrouer, & Straub, 2021).

At the beginning of each flood-regime transition event, the equilibrium topography of the prior regime formed the initial condition for the next interval of landform development. This surface remained until it was either eroded, or buried under new sediment. In this way, the equilibrium topography of the prior regime was juxtaposed with the sediment transport dynamics of the subsequent regime. Gradual events ($\sim T_c$) allowed the landscape to adjust



Geophysical Research Letters



Figure 3. In panel (a), we use a one-way ANOVA to quantify the ratio of time dilation for the control experiments (no change in flood amplitude) versus the positive and negative events (increase or decrease in flood amplitude); that is, $t_{treatment}^*/t_{control}^*$. In panel (b), a one-way ANOVA is used to quantify the difference in phase shift between the control experiment, and the positive and negative treatments; that is, $\phi_{control} - \phi_{treatment}$. Both ANOVAs were fit using a Bayesian framework; here we visualize the posterior estimates of the mean difference, where the color intensity indicates the quantile intervals of the samples. If the posterior distributions overlap unity in panel (a), then estimates of the event duration cannot be statistically distinguished from the control. This is the case for all of the negative treatments (teal). In contrast, all of the positive treatments (brown) plot greater than one, indicating that there is a greater than 90% probability these events are dilated relative to the control experiment. In panel (b), if the posterior distributions group around zero, then they are statistically indistinguishable from the control. Positive treatments all fall below zero, indicating a apparent leading phase shift, whereas negative treatments cluster around zero.

from one state to another while erosion and deposition continuously maintained equilibrium; conversely, abrupt events ($< T_c$) were marked by pronounced, and long-lived disequilibrium (see supplement).

During positive events, increasingly intense floods and mobile channels promoted widespread sediment distribution, which was deposited upon rough topography built by low-amplitude flooding. As a result, there were abundant topographic swales to trap and accumulate sediment, and the flow possessed sufficient energy to distribute sediment into them. In the opposite case (negative events), progressively weaker floods and less mobile channels promoted localized sediment deposition, which accumulated on a smooth surface generated by high-amplitude flooding. Accordingly, sediment deposition was confined to the area adjacent to the channel during negative events (Figure 4).

This hysteresis is the origin of the asymmetric impact on stratigraphic time dilation and phase shift (Figure 3). In the case of positive events, sediment accumulation is spatially widespread, and there is ample storage space, so evidence of the transition event is preserved throughout the basin, resulting in a thicker section than would be expected under steady accumulation. Moreover, the onset of the event is lower in the stratigraphic column than would be expected with steady sediment accumulation, producing an apparent phase lead. In the case of negative events, sediment accumulates in a narrow region around the channel at rates comparable to the control experiment, but some regions of the basin receive no sediment, and many stratigraphic columns contain unconformities spanning the entire event.

5. Reconstructing Time in Dynamic Sedimentary Archives

Our experiments demonstrated that temporary changes in sediment deposition patterns can alter the completeness of the resulting stratigraphic column. By assuming linear time accumulation between dated horizons under these circumstances, geologic events appear to start too early in the stratigraphic column, and unfold over a longer interval of time than they did in reality. The apparent time dilation effect is especially pronounced when the event is abrupt relative to the autogenic dynamics of the sedimentary system.

However, this apparent time distortion is not always present, and under some circumstances, linear time assumptions produced accurate estimates. The key distinguishing factor between these scenarios is the spatial pattern of



Geophysical Research Letters



Figure 4. Red-blue maps shows topography roughness (ridges and swales) as deviations from the overall conical deposit shape, and isopach maps show deposition in grayscale for 10 hr immediately following the transition event. Gray background shows the experimental basin, and the arrow indicates the water and sediment inlet, the shoreline is drawn on the map. The characteristic surface roughness is the standard deviation of the red-blue maps, and panels immediately following each transition. The duration of the preceding transition is given on the *x*-axis in units of T_c . Ranges of equilibrium values for roughness during low and high-intensity flooding are given by shaded regions. Abrupt transitions possess roughness values more typical of the preceding flooding regime, whereas gradual transitions are marked by roughness values in the range of equilibrium values. The two transitions shown in the maps (T_5^+ and T_6^-) are indicated with the teal and brown arrows.

sediment deposition before, during, and after the event. When sediment deposition is localized and confined to a limited area before the event, then during the event becomes progressively widespread, extreme distortion results. When the opposite occurs, distortion is minimized. We propose that sedimentological indicators of localized versus widespread sediment dispersal can be used to identify sections where linear-time assumptions are unlikely to hold, and also to predict the nature of the time distortion in those sections.

Care must be taken when working with small samples or individual cores. Our results reveal that time distortion is predictable on average, but distortion varied substantially from core to core. Small sample sizes lack predictive power; combining age models from many stratigraphic columns will produce the most accurate reconstructions. While acquiring many cores from the same sedimentary system is often impractical, one could in principle use the statistical models presented here to estimate a prediction interval for an individual core and estimate the uncertainty. Ultimately, with additional investigation and careful application, these results could be used to arrive at improved age models for sedimentary strata.

There are many events in the Earth's past where sediment dispersal patterns change as a result of environmental perturbations. For example, in fluvial sedimentary basins where the PETM has been documented, intensified flooding and/or enhanced sediment supply from surrounding mountains commonly resulted in widespread sand

and conglomerate deposition (Barefoot, Nittrouer, Foreman, et al., 2021; Foreman et al., 2012). Relative to the compensation timescale estimated in these basins, the PETM onset is relatively short. Current best estimates indicate that the carbon isotope excursion (CIE) that marks the onset of the PETM could have been as brief as 5 kyr (Kirtland Turner, 2018). In many of the terrestrial basins where the PETM has been studied (e.g., the Bighorn (Foreman, 2014), Piceance (Foreman et al., 2012), and Tremp-Graus (Colombera et al., 2017)), this is anywhere from 1/2 to 1/10th of the basin's compensation timescale (T_c). As a result, the apparent duration of the PETM CIE onset measured from locations like these may be substantially dilated and/or phase shifted (Aziz et al., 2008).

The effects we have described here almost certainly lurk in the sedimentary records of other depositional environments, where biological and geochemical proxies are used to establish the chronology of major events in Earth history. The suite of sedimentological and geological context clues that might indicate time distortion will vary across sedimentary basins and depositional environments. Nonetheless, our experiments suggest a way forward in these circumstances: one should try to distinguish between episodes where sediment deposition is focused in small areas or spread widely across the depositional system.

Marine turbidity currents, for example, can cycle between episodes of relative confinement (where sediment accumulates in a localized area) and unconfinment (where sediment accumulates in large sheets) (Pyles, 2008; Romans et al., 2011). Submarine channels confine themselves by building levees via suspension fallout. Fluctuating sea level can disconnect submarine channels from the upstream continental shelf, and starve the system of suspendible sediment. Ultimately, disconnected channels cannot build levees, and they tend to deposit sediment in unconfined sheets (Romans et al., 2011). The degree of confinement can be inferred after the fact by studying the resulting turbidite deposits (Malkowski et al., 2017). Based on our results, we speculate that within packages with similar degrees of confinement, linear time assumptions will hold. Across the boundaries, linear time likely does not hold for transitions from relatively confined to unconfined sedimentation.

Our experiments suggest a general rule: environmental change that shifts sediment deposition patterns from localized to widespread deposition is likely to distort the preservation of geologic time. However, this need not be the only, or even the most important way that shifting sedimentary processes affect the temporal resolution of stratigraphy. Importantly, there are also feedbacks and interactions between sediment deposition patterns and the biogeochemical markers that stratigraphers use to reconstruct paleoenvironmental histories. It will remain a challenge to untangle multiple different effects from a common cause (Holland, 2016; Peters, 2005, 2006). For example, a change in sea level (Davies et al., 2009) may coincide with the last occurrence of a fossil in the strata, and sea level rise itself also affects sediment localization.

Our experiments isolate and quantify the effect of variable sedimentation rates and corresponding changes in stratigraphic architecture. The results demonstrate that apparent time dilation and phase shifts in sedimentary strata are predictable, and thus suggest that with sufficient information about the autogenic dynamics of an ancient sedimentary system, one can—in principle—separate the impact of changing stratigraphic architecture from environmental impacts on biogeochemical markers hosted in the sediment. Recent theoretical and methodological advances demonstrate how nested hierarchies of autogenic landforms elements interact to produce sedimentary structures. Combined with the experimental results here, these advances establish a framework for deriving quantitative constraints on the characteristic autogenic timescales and lengthscales of ancient sedimentary systems (Ganti et al., 2020; Greenberg et al., 2021; Leary & Ganti, 2020; Lyster et al., 2022). Our results should be used as a first step to leverage quantitative reconstructions of sedimentary autogenics to sharpen the temporal resolution of paleoclimate and paleontological archives when independent age constraints are sparse.

Data Availability Statement

The data that support this study can be downloaded at the following https://doi.org/10.5281/zenodo.8071771. Data are stored as three dimensional arrays, in HDF5 datasets. Analysis was conducted with a mix of R and Python 3, with statistical calculations completed using R and JAGS. Plots were created with Python and R/ggplot2, and all code are available from the same DOI as the data.



Acknowledgments

E.A.B. thanks Robert Mahon, whose help was essential for the experiments. E.A.B. acknowledges support from a Pathfinder Fellowship provided by CUAHSI with support from the National Science Foundation (NSF) Cooperative Agreement No. EAR-1849458. E.A.B. additionally acknowledges support from SEPM via a student Assistance Grant. The authors thank Chris Paola, David Mohrig, Pete Sadler, and Steven Holland, whose helpful comments improved the manuscript.

References

- Abels, H. A., Aziz, H. A., Krijgsman, W., Smeets, S. J. B., & Hilgen, F. J. (2010). Long-period eccentricity control on sedimentary sequences in the continental Madrid Basin (middle Miocene, Spain). *Earth and Planetary Science Letters*, 289(1), 220–231. https://doi.org/10.1016/j. epsl.2009.11.011
- Ajayi, S., Kump, L. R., Ridgwell, A., Kirtland Turner, S., Hay, C. C., & Bralower, T. J. (2020). Evaluation of paleocene-eocene thermal maximum carbon isotope record completeness—An illustration of the potential of dynamic time warping in aligning paleo-proxy records. *Geochemistry*, *Geophysics, Geosystems*, 21(3), e2019GC008620. https://doi.org/10.1029/2019GC008620
- Aswasereelert, W., Meyers, S., Carroll, A., Peters, S., Smith, M., & Feigl, K. (2013). Basin-scale cyclostratigraphy of the green river formation, Wyoming. *GSA Bulletin*, 125(1–2), 216–228. https://doi.org/10.1130/B30541.1
- Aziz, H. A., Hilgen, F. J., Luijk, G. M. V., Sluijs, A., Kraus, M. J., Pares, J. M., & Gingerich, P. D. (2008). Astronomical climate control on paleosol stacking patterns in the upper Paleocene–lower Eocene Willwood Formation, Bighorn Basin, Wyoming. *Geology*, 36(7), 531–534. https://doi.org/10.1130/G24734A.1
- Barefoot, E. A., Nittrouer, J. A., Foreman, B. Z., Hajek, E. A., Dickens, G. R., Baisden, T., & Toms, L. (2021). Evidence for enhanced fluvial channel mobility and fine sediment export due to precipitation seasonality during the Paleocene-Eocene thermal maximum. *Geology*, 50(1), 116–120. https://doi.org/10.1130/G49149.1
- Barefoot, E. A., Nittrouer, J. A., & Straub, K. M. (2021). Non-monotonic floodplain responses to changes in flooding intensity. *Journal of Geophysical Research: Earth Surface*, 126(10), e2021JF006310. https://doi.org/10.1029/2021JF006310
- Bryant, M., Falk, P., & Paola, C. (1995). Experimental study of avulsion frequency and rate of deposition. *Geology*, 23(4), 365–368. https://doi. org/10.1130/0091-7613(1995)023(0365:ESOAFA)2.3.CO;2
- Chadwick, A. J., Lamb, M. P., & Ganti, V. (2020). Accelerated river avulsion frequency on lowland deltas due to sea-level rise. Proceedings of the National Academy of Sciences, 117(30), 17584–17590. https://doi.org/10.1073/pnas.1912351117
- Colombera, L., Arévalo, O. J., & Mountney, N. P. (2017). Fluvial-system response to climate change: The Paleocene-Eocene Tremp group, Pyrenees, Spain. Global and Planetary Change, 157, 1–17. https://doi.org/10.1016/j.gloplacha.2017.08.011
- Davies, N. S., Herringshaw, L. G., & Raine, R. J. (2009). Controls on trace fossil diversity in an early Cambrian epeiric sea: New perspectives from northwest Scotland. *Lethaia*, 42(1), 17–30. https://doi.org/10.1111/j.1502-3931.2008.00130.x
- Davies, N. S., & Shillito, A. P. (2018). Incomplete but intricately detailed: The inevitable preservation of true substrates in a time-deficient stratigraphic record. Geology, 46(8), 679–682. https://doi.org/10.1130/G45206.1
- Davies, N. S., & Shillito, A. P. (2021). True substrates: The exceptional resolution and unexceptional preservation of deep time snapshots on bedding surfaces. Sedimentology, 68(7), 3307–3356. https://doi.org/10.1111/sed.12900
- Esposito, C. R., Leonardo, D. D., Harlan, M., & Straub, K. M. (2018). Sediment storage partitioning in alluvial stratigraphy: The influence of discharge variability. *Journal of Sedimentary Research*, 88(6), 717–726. https://doi.org/10.2110/jsr.2018.36
- Fielding, C. R., Alexander, J., & Allen, J. P. (2018). The role of discharge variability in the formation and preservation of alluvial sediment bodies. Sedimentary Geology, 365, 1–20. https://doi.org/10.1016/j.sedgeo.2017.12.022
- Fike, D. A., Grotzinger, J. P., Pratt, L. M., & Summons, R. E. (2006). Oxidation of the Ediacaran ocean. Nature, 444(7120), 744–747. https://doi. org/10.1038/nature05345
- Foreman, B. Z. (2014). Climate-driven generation of a fluvial sheet sand body at the Paleocene-Eocene boundary in north-west Wyoming (USA). Basin Research, 26(2), 225–241. https://doi.org/10.1111/bre.12027
- Foreman, B. Z., Heller, P. L., & Clementz, M. T. (2012). Fluvial response to abrupt global warming at the Palaeocene/Eocene boundary. *Nature*, 491(7422), 92–95. https://doi.org/10.1038/nature11513
- Foreman, B. Z., & Straub, K. M. (2017). Autogenic geomorphic processes determine the resolution and fidelity of terrestrial paleoclimate records. *Science Advances*, 3(9), e1700683. https://doi.org/10.1126/sciady.1700683
- Gaeuman, D., Schmidt, J. C., & Wilcock, P. R. (2005). Complex channel responses to changes in stream flow and sediment supply on the lower Duchesne River, Utah. *Geomorphology*, 64(3), 185–206. https://doi.org/10.1016/j.geomorph.2004.06.007
- Ganti, V., Hajek, E. A., Leary, K., Straub, K. M., & Paola, C. (2020). Morphodynamic hierarchy and the fabric of the sedimentary record. *Geophysical Research Letters*, 47(14), e2020GL087921. https://doi.org/10.1029/2020GL087921
- Ganti, V., Straub, K. M., Foufoula-Georgiou, E., & Paola, C. (2011). Space-time dynamics of depositional systems: Experimental evidence and theoretical modeling of heavy-tailed statistics. *Journal of Geophysical Research*, 116(F2), F02011. https://doi.org/10.1029/2010JF001893
- Greenberg, E., Ganti, V., & Hajek, E. (2021). Quantifying bankfull flow width using preserved bar clinoforms from fluvial strata. *Geology*, 49(9), 1038–1043. https://doi.org/10.1130/G48729.1
- Halverson, G. P., Hoffman, P. F., Schrag, D. P., Maloof, A. C., & Rice, A. H. N. (2005). Toward a Neoproterozoic composite carbon-isotope record. GSA Bulletin, 117(9–10), 1181–1207. https://doi.org/10.1130/B25630.1
- Hay, C. C., Creveling, J. R., Hagen, C. J., Maloof, A. C., & Huybers, P. (2019). A library of early Cambrian chemostratigraphic correlations from a reproducible algorithm. *Geology*, 47(5), 457–460. https://doi.org/10.1130/G46019.1
- Holland, S. M. (2016). The non-uniformity of fossil preservation. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 371(1699), 20150130. https://doi.org/10.1098/rstb.2015.0130
- Hoyal, D. C. J. D., & Sheets, B. A. (2009). Morphodynamic evolution of experimental cohesive deltas. *Journal of Geophysical Research*, 114(F2), F02009. https://doi.org/10.1029/2007JF000882
- Jarochowska, E., Nohl, T., Grohganz, M., Hohmann, N., Vandenbroucke, T. R. A., & Munnecke, A. (2020). Reconstructing depositional rates and their effect on paleoenvironmental proxies: The case of the lau carbon isotope excursion in Gotland, Sweden. *Paleoceanography and Paleoclimatology*, 35(12), e2020PA003979. https://doi.org/10.1029/2020PA003979
- Jerolmack, D. J., & Sadler, P. (2007). Transience and persistence in the depositional record of continental margins. *Journal of Geophysical Research*, 112(F3), F03S13. https://doi.org/10.1029/2006JF000555
- Keller, G., Sahni, A., & Bajpai, S. (2009). Deccan volcanism, the KT mass extinction and dinosaurs. *Journal of Biosciences*, 34(5), 709–728. https://doi.org/10.1007/s12038-009-0059-6
- Kirtland Turner, S. (2018). Constraints on the onset duration of the Paleocene–Eocene thermal maximum. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences*, 376(2130), 20170082. https://doi.org/10.1098/rsta.2017.0082
- Knoll, A. H., Hayes, J. M., Kaufman, A. J., Swett, K., & Lambert, I. B. (1986). Secular variation in carbon isotope ratios from Upper Proterozoic successions of Svalbard and East Greenland. *Nature*, 321(6073), 832–838. https://doi.org/10.1038/321832a0
- Kominz, M. A., & Bond, G. C. (1990). A new method of testing periodicity in cyclic sediments: Application to the newark supergroup. Earth and Planetary Science Letters, 98(2), 233–244. https://doi.org/10.1016/0012-821X(90)90062-3

- Lane, S. N., Richards, K. S., & Chandler, J. H. (1996). Discharge and sediment supply controls on erosion and deposition in a dynamic alluvial channel. *Geomorphology*, 15(1), 1–15. https://doi.org/10.1016/0169-555X(95)00113-J
- Leary, K. C. P., & Ganti, V. (2020). Preserved fluvial cross strata record bedform disequilibrium dynamics. *Geophysical Research Letters*, 47(2), e2019GL085910. https://doi.org/10.1029/2019GL085910
- Le Guerroué, E., Allen, P. A., Cozzi, A., Etienne, J. L., & Fanning, M. (2006). 50 Myr recovery from the largest negative δ¹³C excursion in the Ediacaran Ocean. *Terra Nova*, *18*(2), 147–153. https://doi.org/10.1111/j.1365-3121.2006.00674.x
- Li, Q., Benson, W. M., Harlan, M., Robichaux, P., Sha, X., Xu, K., & Straub, K. M. (2017). Influence of sediment cohesion on deltaic morphodynamics and stratigraphy over basin-filling time scales. *Journal of Geophysical Research: Earth Surface*, 122(10), 1808–1826. https://doi. org/10.1002/2017JF004216
- Li, Q., & Straub, K. M. (2020). TDB_12_1. SEAD. https://doi.org/10.5967/M03N21GX
- Li, Q., Yu, L., & Straub, K. M. (2016). Storage thresholds for relative sea-level signals in the stratigraphic record. *Geology*, 44(3), 179–182. https://doi.org/10.1130/G37484.1
- Lisiecki, L. E., & Lisiecki, P. A. (2002). Application of dynamic programming to the correlation of paleoclimate records. *Paleoceanography*, 17(4), 1-1–1-12. https://doi.org/10.1029/2001PA000733
- Lyster, S. J., Whittaker, A. C., Hajek, E. A., & Ganti, V. (2022). Field evidence for disequilibrium dynamics in preserved fluvial cross-strata: A record of discharge variability or morphodynamic hierarchy? *Earth and Planetary Science Letters*, 579, 117355. https://doi.org/10.1016/j. epsl.2021.117355
- Malkowski, M. A., Sharman, G. R., Graham, S. A., & Fildani, A. (2017). Characterisation and diachronous initiation of coarse clastic deposition in the Magallanes-Austral foreland basin, Patagonian Andes. *Basin Research*, 29, 298–326. https://doi.org/10.1111/bre.12150
- Meyers, S. R., & Sageman, B. B. (2007). Quantification of deep-time orbital forcing by average spectral misfit. American Journal of Science, 307(5), 773–792. https://doi.org/10.2475/05.2007.01
- Muto, T. (2001). Shoreline autoretreat substantiated in flume experiments. Journal of Sedimentary Research, 71(2), 246–254. https://doi.org/10.1306/091400710246
- Paola, C., Ganti, V., Mohrig, D., Runkel, A. C., & Straub, K. M. (2018). Time not our time: Physical controls on the preservation and measurement of geologic time. Annual Review of Earth and Planetary Sciences, 46(1), 409–438. https://doi.org/10.1146/annurey-earth-082517-010129
- Peters, S. E. (2005). Geologic constraints on the macroevolutionary history of marine animals. Proceedings of the National Academy of Sciences, 102(35), 12326–12331. https://doi.org/10.1073/pnas.0502616102
- Peters, S. E. (2006). Genus extinction, origination, and the durations of sedimentary hiatuses. *Paleobiology*, 32(3), 387-407. https://doi.org/10.1666/05081.1
- Pyles, D. R. (2008). Multiscale stratigraphic analysis of a structurally confined submarine fan: Carboniferous Ross Sandstone, Ireland. AAPG Bulletin, 92(5), 557–587. https://doi.org/10.1306/01110807042
- Rogers, R. R. (1998). Sequence analysis of the upper cretaceous two medicine and Judith River formations, Montana; nonmarine response to the Claggett and Bearpaw marine cycles. *Journal of Sedimentary Research*, 68(4), 615–631. https://doi.org/10.2110/jsr.68.604
- Röhl, U., Westerhold, T., Bralower, T. J., & Zachos, J. C. (2007). On the duration of the Paleocene-Eocene thermal maximum (PETM). Geochemistry, Geophysics, Geosystems, 8(12), Q12002. https://doi.org/10.1029/2007GC001784
- Romans, B., Fildani, A., Hubbard, S., Covault, J., Fosdick, J., & Graham, S. (2011). Evolution of deep-water stratigraphic architecture, Magallanes Basin, Chile. Marine and Petroleum Geology, 28(3), 612–628. https://doi.org/10.1016/j.marpetgeo.2010.05.002
- Sadler, P. M. (1981). Sediment accumulation rates and the completeness of stratigraphic sections. *The Journal of Geology*, 89(5), 569–584. https://doi.org/10.1086/628623
- Sadler, P. M., & Strauss, D. J. (1990). Estimation of completeness of stratigraphical sections using empirical data and theoretical models. *Journal of the Geological Society*, 147(3), 471–485. https://doi.org/10.1144/gsjgs.147.3.0471
- Salter, G., Paola, C., & Voller, V. R. (2018). Control of delta avulsion by downstream sediment sinks. Journal of Geophysical Research: Earth Surface, 123(1), 142–166. https://doi.org/10.1002/2017JF004350
- Schoene, B., Eddy, M. P., Samperton, K. M., Keller, C. B., Keller, G., Adatte, T., & Khadri, S. F. R. (2019). U-Pb constraints on pulsed eruption of the Deccan Traps across the end-Cretaceous mass extinction. *Science*, 363(6429), 862–866. https://doi.org/10.1126/science.aau2422
- Schoene, B., Samperton, K. M., Eddy, M. P., Keller, G., Adatte, T., Bowring, S. A., et al. (2015). U-Pb geochronology of the Deccan Traps and relation to the end-Cretaceous mass extinction. *Science*, 347(6218), 182–184. https://doi.org/10.1126/science.aaa0118
- Schulte, P., Alegret, L., Arenillas, I., Arz, J. A., Barton, P. J., Bown, P. R., et al. (2010). The Chicxulub asteroid impact and mass extinction at the cretaceous-Paleogene boundary. *Science*, 327(5970), 1214–1218. https://doi.org/10.1126/science.1177265
- Schumer, R., Jerolmack, D., & McElroy, B. (2011). The stratigraphic filter and bias in measurement of geologic rates. *Geophysical Research Letters*, 38(11), L11405. https://doi.org/10.1029/2011GL047118
- Sharpton, V. L., Brent Dalrymple, G., Marín, L. E., Ryder, G., Schuraytz, B. C., & Urrutia-Fucugauchi, J. (1992). New links between the Chicxulub impact structure and the Cretaceous/Tertiary boundary. *Nature*, 359(6398), 819–821. https://doi.org/10.1038/359819a0
- Straub, K. M., & Foreman, B. Z. (2018). Geomorphic stasis and spatiotemporal scales of stratigraphic completeness. *Geology*, *46*(4), 311–314. https://doi.org/10.1130/G40045.1
- Straub, K. M., Paola, C., Mohrig, D., Wolinsky, M. A., & George, T. (2009). Compensational stacking of channelized sedimentary deposits. Journal of Sedimentary Research, 79(9), 673–688. https://doi.org/10.2110/jsr.2009.070
- Syvitski, J. P. M., & Saito, Y. (2007). Morphodynamics of deltas under the influence of humans. *Global and Planetary Change*, 57(3), 261–282. https://doi.org/10.1016/j.gloplacha.2006.12.001
- Tipper, J. C. (2015). The importance of doing nothing: Stasis in sedimentation systems and its stratigraphic effects. *Geological Society, London, Special Publications, 404*(1), 105–122. https://doi.org/10.1144/SP404.6
- Trampush, S. M., & Hajek, E. A. (2017). Preserving proxy records in dynamic landscapes: Modeling and examples from the Paleocene-Eocene Thermal Maximum. *Geology*, 45(11), 967–970. https://doi.org/10.1130/G39367.1
- Trauth, M. H. (2014). A new probabilistic technique to build an age model for complex stratigraphic sequences. *Quaternary Geochronology*, 22, 65–71. https://doi.org/10.1016/j.quageo.2014.03.001
- Wang, Y., Straub, K. M., & Hajek, E. A. (2011). Scale-dependent compensational stacking: An estimate of autogenic time scales in channelized sedimentary deposits. *Geology*, 39(9), 811–814. https://doi.org/10.1130/G32068.1
- Westerhold, T., Röhl, U., McCarren, H. K., & Zachos, J. C. (2009). Latest on the absolute age of the Paleocene–Eocene Thermal Maximum (PETM): New insights from exact stratigraphic position of key ash layers +19 and -17. *Earth and Planetary Science Letters*, 287(3), 412–419. https://doi.org/10.1016/j.epsl.2009.08.027

- Wilf, P., Johnson, K. R., & Huber, B. T. (2003). Correlated terrestrial and marine evidence for global climate changes before mass extinction at the Cretaceous–Paleogene boundary. *Proceedings of the National Academy of Sciences*, 100(2), 599–604. https://doi.org/10.1073/ pnas.0234701100
- Williams, G. E., & Schmidt, P. W. (2018). Shuram–Wonoka carbon isotope excursion: Ediacaran revolution in the world ocean's meridional overturning circulation. *Geoscience Frontiers*, 9(2), 391–402. https://doi.org/10.1016/j.gsf.2017.11.006
- Yu, L. (2002). The Huanghe (Yellow) river: A review of its development, characteristics, and future management issues. Continental Shelf Research, 22(3), 389–403. https://doi.org/10.1016/S0278-4343(01)00088-7
- Zachos, J. C., Lohmann, K. C., Walker, J. C. G., & Wise, S. W. (1993). Abrupt climate change and transient climates during the Paleogene: A marine perspective. *The Journal of Geology*, 101(2), 191–213. https://doi.org/10.1086/648216
- Zachos, J. C., Röhl, U., Schellenberg, S. A., Sluijs, A., Hodell, D. A., Kelly, D. C., et al. (2005). Rapid acidification of the ocean during the paleocene-eocene thermal maximum. *Science*, 308(5728), 1611–1615. https://doi.org/10.1126/science.1109004

References From the Supporting Information

- Kim, W., Paola, C., Swenson, J. B., & Voller, V. R. (2006). Shoreline response to autogenic processes of sediment storage and release in the fluvial system. *Journal of Geophysical Research*, 111(F4), F04013. https://doi.org/10.1029/2006JF000470
- Straub, K. M., & Esposito, C. R. (2013). Influence of water and sediment supply on the stratigraphic record of alluvial fans and deltas: Process controls on stratigraphic completeness. *Journal of Geophysical Research: Earth Surface*, 118(2), 625–637. https://doi.org/10.1002/jgrf.20061 Strong, N., & Paola, C. (2008). Valleys that never were: Time surfaces versus stratigraphic surfaces. *Journal of Sedimentary Research*, 78(8),

579–593. https://doi.org/10.2110/jsr.2008.059